Author's response to reviews

Title: Effects of group medical appointments on quality of life and cost-effectiveness for patients with chronic neuromuscular disease. Study protocol of a randomized controlled trial.

Authors:

Femke M Seesing (f.seesing@neuro.umcn.nl)
Gea Drost (g.drost@neuro.umcn.nl)
Gert-Jan van der Wilt (g.vanderwilt@ebh.umcn.nl)
Baziel GM van Engelen (b.vanengelen@neuro.umcn.nl)

Version: 3 Date: 10 August 2011

Author's response to reviews:

Author list: the list of authors in the manuscript is written exactly as they are in the submission system, both in style and order. Titles are removed.

Tables are now numbered in ascending numerical order: 1, 2 and 3, and the references to them in the text.

Structure: The abbreviations section now comes before the competing interests section.

Tables: all vertical lines are removed.